Author’s response to reviews

Title: Effectiveness of proactive telephone counselling for smoking cessation in parents: Study protocol of a randomized controlled trial

Authors:

Kathrin Schuck (k.schuck@bsi.ru.nl)
Roy Otten (r.otten@bsi.ru.nl)
Marloes Kleinjan (m.kleinjan@bsi.ru.nl)
Jonathan B. Bricker (jbricker@fhcrc.org)
Rutger C. M. E. Engels (r.engels@bsi.ru.nl)

Version: 5 Date: 13 September 2011

Author’s response to reviews: see over
Dear Sir / Madam,

Thank you for accepting our manuscript ‘Effectiveness of proactive telephone counselling for smoking cessation in parents: Study protocol of a randomized controlled trial’ for publication in *BMC Public Health*.

In response to the editorial request we have made the following changes to the manuscript:

- Figure 1 is displayed on a single page
- Figure title was included
- Title page: ‘adress’ was changed into ‘address’
- Abstract: bold heading of trial registration was removed

Yours sincerely,

Kathrin Schuck
Behavioural Science Institute
Radboud University Nijmegen
PO Box 9104
6500 HE Nijmegen, The Netherlands
tel.: 00 31 243612705
fax.: 00 31 243612776
e-mail.: k.schuck@bsi.ru.nl